Author’s response to reviews

Title: A Decade of Road Traffic Fatalities among the Elderly in North-West Iran

Authors:

Homayoun Sadeghi-Bazargani (homayoun.sadeghi@gmail.com)
Bahram Samadirad (samadirabahram@gmail.com)
Farnaz Moslemi (farnazmoslemitabriz@gmail.com)

Version: 5 Date: 20 Sep 2017

Author’s response to reviews:

Dear Editor,

Thanks for the decision as minor revision on our manuscript entitled “A Decade of Road Traffic Fatalities among the Elderly in North-West Iran”. We revised it as requested. Following is the response to the comments.

Comment:

Editor Comments (Teresa Senserrick):

Thank you for attending to the revisions. I am happy to accept with the words "considering EMS limitations" deleted from line 26 on page 11 (you did not establish this by your study so must not state it as a conclusion in your Discussion).

Response:

Corrected as requested removing the words "considering EMS limitations" in revised manuscript.

Best Regards,

Corresponding author